Author’s response to reviews

Title: Measurement properties of adult quality of life measurement instruments for eczema: protocol for a systematic review

Authors:

Christian J Apfelbacher (christian.apfelbacher@klinik.uni-regensburg.de)
Daniel Heinl (daniel.heinl@stud.uni-regensburg.de)
Cecilia AC Prinsen (c.prinsen@vumc.nl)
Stefanie Deckert (Stefanie.Deckert@uniklinikum-dresden.de)
Robert Ofenloch (Robert.Ofenloch@med.uni-heidelberg.de)
Rosemary Humphreys (RHNumphreys@aol.com)
Tracey Sach (t.sach@uea.ac.uk)
Sarah Chamlin (schamlin@luriechildrens.org)
Jochen Schmitt (jochen.Schmitt@uniklinikum-dresden.de)
Joanne Chalmers (joanne.Chalmers@nottingham.ac.uk)

Version: 4
Date: 27 March 2015

Author’s response to reviews:

Dear Editors,

as requested we have now included the PROSPERO registration number also at the end of the abstract, have changed reference 6 in the methods (now relating to PRISMA-P) and have explained each author’s contribution individually.

Sincerely yours,

Christian Apfelbacher